Author’s response to reviews

Title: A randomized home-based childhood obesity intervention delivered by patient navigators

Authors:

Lourdes WH Yun (lourdes.yun@dhha.org)
Richard E Boles (richard.boles@ucdenver.edu)
Matthew A Haemer (matthew.haemer@ucdenver.edu)
Shanna Knierim (shanna.knierim@ucdenver.edu)
Linda M Dickinson (miriam.dickinson@ucdenver.edu)
Heather Mancinas (heathermancinas11@gmail.com)
Simon J Hambidge (simon.hambidge@dhha.org)
Arthur J Davidson (arthur.davidson@dhha.org)

Version: 2
Date: 6 May 2015

Author’s response to reviews:

There are 2 cover letters attached. Please use the one entitled: "CoverLetter_20150506. I could not delete the prior cover letter (also attached)."